Working Paper 426

GLOBAL CRISIS, ENVIRONMENTAL VOLATILITY AND EXPANSION OF THE INDIAN LEATHER INDUSTRY

Anup Kumar Bhandari

March 2010

Working Papers can be downloaded from the Centre's website (www.cds.edu)

GLOBAL CRISIS, ENVIRONMENTAL VOLATILITY AND EXPANSION OF THE INDIAN LEATHER INDUSTRY

Anup Kumar Bhandari

March 2010

The author likes to thank the seminar participants at the Jadavpur University, Calcutta, University of Burdwan, West Bengal and Institute of Rural Management Anand, Gujarat and an anonymous referee for their comments and suggestions on an earlier draft of the paper. The usual disclaimer applies.

ABSTRACT

Indian leather industry has massive potential for generating employment and achieving high export-oriented growth. However, the on-going global economic slowdown and the wide erratic behaviour of the overall weather condition particularly in the Europe pose both threat (of market loss) and opportunity (to gain some unanticipated demand in the market) before it. On the other hand, its economic performance has not been assessed much till date. The present paper attempts to fill in this gap and makes some suggestions regarding the expansion of the industry by examining technical efficiency (TE) of individual leather producing firms for some selected years since the early-1980s. Analysing the industry's firm-level data through the Data Envelopment Analysis the paper observes a significant positive association between a firm's size and its TE, but no such conclusive relation between a firm's age and TE. It also finds significant variation in TE across firms in different groups of states as well as under different organisational structures. Although, non-availability of panel data does not allow one to assess the trend of the performance of the Indian leather firms properly, the average firm-level TE, however, seems to be on an increasing path, except for downswing in some years. On a totality, analysing the relevant supply side factors the paper proposes the policy makers to go forward in expanding the industry particularly keeping India's severe unemployment problem, of both skilled as well as unskilled labour forces, in mind.

Key Words: Leather industry; Data Envelopment Analysis, Technical Efficiency, Scale Efficiency, Returns to Scale.

JEL Classification No: D24, L67, R38.

1. Introduction

The leather industry occupies a place of prominence in the Indian economy in view of its massive potential for employment, growth and exports. In fact, backed by a strong raw-material base and a large reservoir of traditionally skilled and competitive labour force, the Indian leather industry has made significant strides during the past two decades.¹ Not only that, this industry has undergone a dramatic transformation from a mere exporter of raw materials (like tanned hides and skins) in the 1960s to that of value added finished products from the 1970s. Policy initiatives taken by the Government of India (GOI) since 1973 have been quite instrumental to such a transformation.

The structure of the Indian leather industry is quite interesting. It is spread in different segments namely tanning and finishing, footwear and footwear components, leather garments, leather goods including saddlery and harness etc.² The industry uses primarily indigenous natural resources with little dependence on imported resources. Hides and skins are the basic raw materials for the leather industry, which originate from the source of livestock. India has a very large share of the world bovine

¹ For instance, export of leather and leather manufacturers (including leather footwear, leather travel goods and leather garments) went up from US \$59 million in 1960-61 to US \$493 million in 1980-81 and thereafter to US \$1449 million in 1990-91 and further to US \$2323 million in 2004-05 (Government of India, 2004-05).

² Detailed discussion on the organisational structure of the industry starting from the stage of collection of raw materials to that of marketing of finished and semi-finished products can also be found in Banerjee and Nihila (1999), Mohapatra and Srivastava (2002) and Singh (2007).

animal population.³ Further, an overwhelming proportion of the total production of this industry comes from the unorganised sector, i.e., small scale, cottage and artisan sector. The major production centres are spread over selected areas in a few states, e.g., selected places in Tamil Nadu, Kolkata in West Bengal, Kanpur and Agra in Uttar Pradesh, Jalandhar in Punjab and Delhi. And the major export market for Indian leather goods is Germany, with an oftake of about 25 percent of India's domestic production, followed by the USA, the UK, France and Italy. The important export items are leather handbags, footwear and leather garments.

Official policies/programmes undertaken to facilitate the growth of the leather industry include de-reservations of 11 items (particularly semi-finished hides and skins, leather shoes and leather accessories for leather industry) in 2001, abolition of the license system in case of manufacture of most of the leather items. Some items are still reserved for exclusive manufacture by the small-scale sector, but non-small scale units can also obtain approval for the manufacture of these items provided they meet an export obligation of 50 per cent of their annual production.⁴

³ For instance, India's share in the world bovine animal population in 2005 was the highest (about 19%), followed by Brazil (13%), China (9%) and the USA (6%).

In addition, a number of leather development programmes have been initiated 4 in the recent past. A UNDP assisted National Leather Development Programme (NLDP Phase I) was carried out from 1992 to 1998 to upgrade the training systems for design and manufacture of footwear, garments and leather goods and its second phase - called the Small Industries Development and Employment Programme (SIDE - NLDP) - from 1998 to 2002, with a view to promoting poverty alleviation and building linkages between the organised and unorganised sectors. To complement the above mentioned programmes a new plan scheme titled Indian Leather Development Programme (ILDP) started operation in 1992 to bridge critical gaps in infrastructure for integrated development of this industry, to undertake investment/trade development activities and build up an information base for leather industry. Productivity improvement programmes have also been launched for improving the manufacturing processes of footwear in the organised sector. A scheme for tannery modernisation was launched under ILDP in 2000 to provide the much needed financial help to the Indian tanneries for adoption of more efficient and cleaner process technologies for improving their performance in terms of productivity and pollution control.

Governments at both the central as well as state levels are in a way to promote formation of more leather manufacturing clusters in different parts of the country to get reap of the benefits from such clustering in both the cost reducing and quality enhancing dimensions. For instance, the Kerala Industrial Infrastructure Development Corporation (KINFRA) is setting up a footwear park at Vazhakkat, near Ramanattukara, in Malappuram district (Business Line, October 7, 2009). The Cabinet Committee on Economic Affairs, Government of India recently gave its nod for development of a leather park under Indian Leather Development Programme (ILDP) and earmarked INR3000 million for it. The aim of the sub-scheme is to provide the industry with infrastructure facilities for setting up leather units across product categories and to attract large domestic joint venture and foreign investments into the Indian leather industry. A leather park set up under this sub-scheme would cover all sectors of leather industry – tannery, all products categories and leather machinery (Business Line, October 23, 2009).⁵

Some intrinsic problems affect the leather firms also. Prior to its detailed discussion we can briefly present the nature of the production

⁵ Beside these Government initiatives and the Central Leather Research Institute (CLRI), Chennai, a central hub in Indian leather sector with direct roles in education, research, training, testing, designing, forecasting, planning, social empowerment and leading in science and technology relating to leather, some other agencies are stretching their helping hands towards the workers by providing necessary training that a world class modern production unit of the industry needs. For instance, one month (January 1-January 31, 2009) training programme for 26 unemployed workers was conducted by the technical expert of Footwear Design & Development Institute (FDDI), Noida at the Florence Shoe Company Private Limited (with its up to date modern management and technical know-how), Ambur, Chennai under the 'Workers Training Programme' of the Department of Industrial Policy & Promotion (DIPP). In this context Mr. Rajeev Lakhara, IRS, Managing Director, FDDI said that the manpower crunch in the Indian leather industry is seen as a major deterrent especially when foreign entities look at India as a region with low production cost. Hence, it's time to train unemployed workers which will be inducted in the factories and play a vital role in the growth path of the leather sector (FDDI Weekly Newsletter, February 13, 2009).

chain of the industry. The entire production process can be sub-divided into three segments: leather tanning (i.e., raw to semi-finished stage), leather finishing (i.e., semi-finished to finished stage) and making of leather goods/products (i.e., the final stage). Each of these segments has six to eight processes. Tanning is basically converting animal hides (outer covering of cow, buffalo and calf) and skins (outer covering of sheep and goat) into leather. The quality of leather products critically depends on the quality of tanning. However, the activities relating to the processing of leather generate pollution, particularly in the tanning⁶ and finishing stages of the production chain and hence, the firms have to bear increasing costs of production for undertaking pollution abating activities and/or relocating their establishments. There are however, some *favourable* factors also. As we have already explored to some extent in footnote 5, major world tanning firms are in the process of shifting their manufacturing base to developing countries due to high wage levels and strict environmental norms in the developed countries. Factors such as sufficient availability of raw leather and cheap skilled labour with their long experience in the technical know-how of production and processing of leather items-all work in India's favour. Further, given that the Indian leather industry is still dominated by household and smallscale sectors, more corporate presence may enhance the possibility of turning out quality leather products at smaller unit cost. All this presents a large scope of expansion before the Indian leather industry.

So, given that there is a large scope of expanding leather production and exporting such products abroad, the question arises whether the present structure of this industry is such as to facilitate such an expansion and, if not, what additional measures are called for. All this needs a thorough examination of its production structure and related

6

The leather tanning industry has been designated as 'hazardous industry' under the Factories Act, 1948.

features – features focusing on the economic performance of the Indian leather firms. For instance, one may like to know how efficient the different leather firms are technically and/or relative to their scale of operation. Such questions have become important as policy makers are gradually taking necessary steps so as to make the overall productive environment more competitive by removing detrimental policy instruments (introduced earlier to draw private investments in the desired directions) during the process of economic reforms which the economy is going through and which calls for an enhanced efficiency of firms for their growth or even mere survival.

Given the global economic meltdown in the backdrop the present situation, however, creates some hesitation before the policy makers concerning the issue of expansion of the Indian leather industry, at least in the present short run. To be specific, the proposed prospect of the Indian leather industry, as we have already mentioned, largely lies in its export earning potential with West European developed countries and the USA being its major export market. But the on-going global economic crisis affects these countries very badly. Rapidly increasing unemployment in these countries lowers purchasing power of the mass and, as an obvious consequence, overall demand for consumer goods is diminishing with that for leather products is not an exception. According to the industry representatives the Indian leather industry is looking at a 20 per cent drop in output following the slowdown in demand in the international market (Business Line, January 29, 2009). The situations even further worsen for Chinese intervention into the European market since China is facing a more intense slowdown because of its focus on the US market and trying to wean away India's European customers (Business Line, January 31, 2009). The Government of India already announced a package valued INR3.25 billion for the leather and textile export sectors (Business Line, February 27, 2009). Government declared a further 2 per cent incentive for exports of garments and all leather items to the European Union and

the US which would be effective from April 1, 2009 and shall remain available till end-September 2009 (*Business Line*, March 1, 2009). Although the quantitative volume of these packages was whether determined by some rigorous estimation or much more induced by the then forthcoming general election of the country is a debatable issue, the need for some support programme is undoubted.

On the other hand, one of the important factors affecting the demand (in a particular time) for leather goods, particularly the leather garments is the prevailing weather condition. Volatility of the overall climatic condition worldwide, as we are experiencing now probably due to global warming, brings some unanticipated blessing to the Indian leather industry in the present crisis-hit unfavourable global market. "The weather god may just be working in favour of the leather goods industry in Europe this year in offsetting recession. The heavy snowfall and cold recorded in most of European countries this year has resulted in a considerable sale of shoes and leather garments, despite recession", according to Mr. Mario Pucci, International Relations Director, National Association of Italian Manufacturers of Footwear, Leather Goods and Tanning Machinery, ASSOMAC (Business Line, February 7, 2009). He said at a seminar in Kolkata on February 6, 2009 that the Indian leather manufacturers may also find a larger market in Italy and other European countries this year as most stores there are already running short of stocks (due to recession).

So, current overall global scenario poses both opportunity and threat before the Indian leather industry. It can be easily understood that these two opposite factors may affect the Indian leather industry from the demand side. The objective of this paper is to examine the supply side factors in this connection and to draw necessary policy conclusion about to which direction the industry should be driven, – expansion or contraction, by the policy makers in the face of the ongoing global economic peculiarities. For that, we have to examine rigorously some

selected features of the production behaviours of the Indian leather industry, particularly those related to the firm size – technical efficiency relationship and the region of the production technology where majority of the firms are situated – whether it is the increasing returns to scale (IRS) or the diminishing returns to scale (DRS) segment of the frontier. Age of a firm may also affect its technical efficiency significantly. We also want to examine the overall efficiency level of the Indian leather firms and see whether it is improving over time or not.

Size and age, among others, are two important variables emphasized in the literature those can affect efficiency of a firm. A large firm may have an easier access to cheaper or superior quality of inputs which makes it more productive. Moreover, it may exploit economies of scale, broaden its scope and also find it easier to obtain all the requisite official approvals. These features, by making its operation more effective, allow it to perform better relative to the smaller ones (Penrose, 1959). However, the relation between firm's age and performance may not be so clear cut. One line of argument is that older firms display superior performance as they are more experienced, able to reap the benefits of learning and free from the hazards which newcomers usually face (Stinchcombe, 1965). There are, however, counter arguments as well viz. that the older firms are prone to inertia, lacking adequate flexibility to adapt rapidly to the changing economic circumstances which younger and more agile firms can do much more quickly and efficiently (Marshall, 1920). Also, younger firms may have assets/plants of later generations embodying advanced technology and also younger, more recently educated workers and hence may have higher efficiency (Lall and Rodrigo, 2001).

However, the paper is organised as follows. Section 2 briefly reviews the concerned literature and Section 3 discusses the methodology followed in the paper. Section 4 describes our data set and empirical findings and Section 5 concludes the paper and suggests some policy conclusions. Appendix presents some more results.

2. A Brief Review of Literature

In this connection we may briefly review the literature grown so far, both at the micro and at the aggregative level. It may be noted that we have come across only a few studies in the context of Indian leather industry. So we briefly review studies on technical efficiency (TE) of firms in the context of other industries or the entire industrial sector in India. Some studies use data collected through surveys specifically designed for this purpose (e.g., Little, Mazumdar and Page, 1987; Page, 1984). A number of studies are concerned with estimating and explaining variations in TE in only the small-scale industrial units by fitting either a deterministic or a stochastic production frontier (e.g., Bhavani, 1991; Goldar, 1985; Neogi and Ghosh, 1994; Nikaido, 2004; Ramaswamy, 1994). These studies, however, use data relating to prereform period. For instance, using the data thrown up by the first Census of Small Scale Industrial Units, 1973, Bhavani (1991) estimates TE's of firms in four metal product groups by fitting a (deterministic) translog production frontier with three inputs, capital, labor and materials and observes a very high level of average efficiency across the four groups. Similarly, on the basis of the data collected under the Second All India Census of Small scale Industrial Units, 1987-88, Nikaido (2004) fits a single stochastic production frontier, considering firms under all of the two-digit industry-groups, using intercept dummies to distinguish different industry groups. He finds little variation in TE across industry groups and a high level of average TE in each industry group. Neogi et al (1994), using industry-level summary data for the years 1974-75 to 1987-88, observes TE's of firms under different industries to be falling over time. The studies by Goldar, Renganathan and Banga (2004), Lall et al (2001) and Mukherjee and Ray (2004), however, relate to the post-reform era. Using the panel data from the Prowess data base of the Centre for Monitoring Indian Economy (CMIE) for 63 firms in the engineering industry for ten years in the 1990's, Goldar et al (2004)

fits a translog stochastic production frontier and finds mean TE of foreign firms to be higher than that of domestically owned firms, but no statistically significant variation in mean TE across public and private sector firms among the latter group. At the second stage they attempt to explain variation in TE in terms of some economic variables like export and import intensity, degree of vertical integration etc. Lall et al (2001) fits a translog stochastic frontier to the plant level data for 1994 on each of four product groups, viz., leather product, motor vehicles, machine tools and electronics and computers. For instance, for the leather product group it finds the distribution of plant level efficiency to be highly negatively skewed with the mean value at 0.44 and further that such efficiency is positively related to the plant's energy use, but not much affected by its age or size. Mukherjee et al (2004) analyse the Annual Survey of Industries (ASI) data for the years 1986-87 through 1999-2000 and finds no major change in the efficiency ranking of individual states as well as no convergence in the distribution of TE across states in post-reform period. On the basis of the firm-level ASI data for a number of years Bhandari and Maiti (2007) and Bhandari and Ray (2007) attempt to measure TE of the Indian textiles firms and find considerable room for improving their efficiency. Bhandari and Maiti (2007) fits a translog stochastic frontier to the data for each of the selected years and finds that the firm-level TE, ranging between 68 to 84 per cent over the sample years, varies positively with farm size and negatively with farm age in any given year and that the public sector firms are relatively less efficient than their private sector counterparts. Applying data envelopment analysis method to the same set of data, Bhandari and Ray (2007) computes firm-level TE and arrives at the same set of conclusions except for no strong evidence of effect of a firm's age on its TE. Bhandari (2009) have done some works on Indian leather industry also.

3. Model⁷ for Measuring Efficiency: Data Envelopment Analysis (DEA)

The non-parametric DEA method introduced by Charnes, Cooper and Rhodes (1978) and further generalised by Banker, Charnes and Cooper (1984) requires no parametric specification of the production frontier. On the basis of a sample of observed input-output data on a given set of producing units, it makes a few assumptions about production technology in order to obtain a production possibility set relevant for the observed units.⁸ The corresponding *frontier*, basically a piece-wise linear surface over the data points, is then constructed by the solution of a sequence of linear programming problems – one for each unit in the sample. It then yields, as a by-product, the extent of technical inefficiency of each unit, i.e., the distance between the observed data point corresponding to the unit and the frontier so constructed. For instance, it will show how much proportional increase of the observed output vector of a unit (firm, in our study) were feasible, given the latter's observed use of various inputs. This proportion is then used to get a measure of (an *output – oriented*) technical efficiency of the firm. We describe the approach below for the case of single-output firms (say, N in number), although the method is a general one applicable to the case of multi-product firms as well.

⁷ Two other alternative methods, widely known as the stochastic frontier analysis (SFA) method and method based on the random coefficient regression model may also be applied to this study. However, we do not use the latter while we use SFA only partially in our study.

⁸ The *assumptions* about the production technology that are made in this method are as follows: (a) All observed input-output bundles are feasible; (b) the production possibility set is convex implying that given a set of *N* feasible input-output bundles, *any* weighted average of these *N* input bundles can produce the same weighted average of the corresponding output bundles and (c) any input or output is *freely* disposable. These assumptions enable one to construct a production possibility frontier on the basis of the observed inputs-output bundles of a given set of firms, following the DEA method.

Let the firm *i* be observed to produce Y_i quantity of the output by using the input bundle X_i , an *m*-component (column) vector of inputs (the *j*th element of X_i is taken to be zero, if the *i*th firm does not use the input). The DEA method seeks to construct a frontier on the basis of the observations on inputs and outputs of all the *N* firms, by solving a set of *N* linear programmes, one for each firm. The problem for the firm *k* is to find a scalar φ and *N* number of λ_{ik} 's which solve the following linear programme.

 $\begin{array}{ll} \left(P_k^O \right) & Maximise \ \varphi \\ \mbox{subject to} & (i) \ \sum_{i=1}^N Y_i \ \lambda_{ik} \ \ge \ \varphi \ Y_k \ , \ (ii) \ \sum_{i=1}^N X_i \ \lambda_{ik} \ \le \ X_k \ , \ (iii) \ \sum_{i=1}^N \lambda_{ik} \ = \ 1, \ \ \mbox{and} \\ (iv) \ \lambda_{ik} \ \ge 0 \ \ \mbox{for all} \ i = 1, ..., N \ . \end{array}$

Let (P_k^O) has an optimal solution, say $[\varphi_k^*, (\lambda_{1k}^*, ..., \lambda_{Nk}^*)]$. The optimal value φ_k^* then gives the maximum possible *proportional increase* in output that could be achieved by the k^{th} firm, with their input quantities being held constant.⁹ Thus the technical efficiency of the k^{th} firm relative to the frontier (TE_k^O) is given by $TE_k^O = 1/\varphi_k^*$.¹⁰

⁹ Note that $[\varphi_k = 1 \text{ and } \{\lambda_{kk} = 1 \& \lambda_{ik} = 0 \text{ for all } i \neq k\}]$ is a feasible solution to the above problem. Hence, the optimal value, φ_k^* , will be greater than or equal to one. In case of multi-product firms, Y_i and Y_k in the first inequality will each be a vector (of appropriate dimension), and not a scalar as is being assumed here. The meaning of the term 'proportional increase in output' (vector) will be quite clear then.

¹⁰ Observe that the DEA programme mentioned here is for the (general) variable returns to scale (VRS) technology. If we are to obtain technical efficiency of a firm under the constant returns to scale (CRS) specification of the technology, we have to solve the same LP problem without the constraint (3) and, therefore, such efficiency under the latter will be no more than that under the earlier technological specification.

Similarly, to obtain technical efficiency of the k^{th} firm in an inputoriented way we have to solve the programme

$$\begin{pmatrix} P_k^I \end{pmatrix} \qquad Minimise \quad \theta$$
subject to (i) $\sum_{i=1}^N Y_i \lambda_{ik} \ge Y_k$, (ii) $\sum_{i=1}^N X_i \lambda_{ik} \le \theta X_k$,
(iii) $\sum_{i=1}^N \lambda_{ik} = 1$, and (iv) $\lambda_{ik} \ge 0$ for all $i = 1, ..., N_k$
be (input oriented) technical efficiency of the k^{th} firm relative to

and the (input-oriented) technical efficiency of the k^{th} firm relative to the frontier is given by $TE_k^I = \theta_k^*$ where θ_k^* being the optimal solution to the problem P_k^I .



These concepts will be clearer if we discuss with the help of a diagram. Let ATBC (in Figure 1) be the production frontier (exhibiting VRS technology with other usual desirable properties). An (outputoriented) measure of technical efficiency (TE) of firm F, as defined to be the ratio of actually produced amount of output to the frontier level of output for the given level of input used by this firm, is given by $\frac{FX_1}{BX_1} = \frac{FX_1/OX_1}{BX_1/OX_1}$ which is equal to the ratio of productivity, as

defined to be the amount of output per unit of input used, at the point F to that at the point B. Note that TE is identical (and equals to unity) at all points on the frontier, but productivity is not. It is easy to see that productivity is the highest at the point T among all feasible points (i.e., those lie within the production possibility set). Hence, OX* is the most productive scale size (MPSS) (a la Banker, 1984) in the above diagram. Output-oriented scale efficiency of a firm is defined to be the ratio of productivity at its (output-oriented) projection on to the frontier to that at the MPSS. Similarly, input-oriented measure of scale efficiency of a firm is the ratio of productivity at its (input-oriented) projection on to the frontier to that at the MPSS. In other words, scale efficiency is a measure of relative productivity of a firm with respect to productivity at the MPSS, if the firm becomes able to eliminate its technical inefficiency in production and, therefore, naturally it lies between 0 and 1.¹¹ So, scale efficiency of any firm lies on the vertical line BX_1 is, $\frac{BX_1/OX_1}{TX^*/OX^*}$ which is the ratio of productivity at point B to that at point

T and scale efficiency of any firm lies on the horizontal line B_1F is the ratio of productivity at the point B_1 to that at point T. But, productivity at point T is equivalent to that of the hypothetical firms lie at point D and D_1 . Although, these points are not feasible under the VRS technology, they are on the graph of the CRS technology. Thus,

¹¹ Note that scale efficiency does not state anything about the actual scale of production relative to the MPSS, in the sense that one cannot say whether the firm actually practicing more or less than the MPSS, by simply observing its scale efficiency score.

$$\frac{BX_1/OX_1}{TX^*/OX^*} = \frac{BX_1/OX_1}{DX_1/OX_1} = \frac{BX_1}{DX_1} = \frac{FX_1/DX_1}{FX_1/BX_1}$$
 and similarly we can

show that the ratio of productivity at B₁ to that at D₁ is equal to the ratio $\frac{Y_1D_1/Y_1F}{Y_1B_1/Y_1F}$. So, scale efficiency of a firm is the ratio of its TE under the CRS technology to that under the VRS technology.

Note that the frontier shown in the above diagram exhibits all the three types of local returns to scale property along its different segments.¹² Through its AT (TC) segment, increasing (diminishing) returns to scale holds and at the point T local constant returns to scale holds. Clearly, a firm uses less (more) than OX^{*} amount of inputs have to increase (decrease) its production scale in order to attain MPSS. We say that a firm follows IRS (DRS) technology if its projection on to the frontier touches the frontier in its IRS (DRS) segment. Since, there may be some firms whose output-oriented projection touches the frontier in its IRS segment, finally we take *only* those firms to follow IRS (DRS) technology whose output-oriented (input-oriented) projection is on the IRS (DRS) segment of the frontier.¹³

¹² See Banker (1984), Banker et al (1984) and Färe et al (1985) to obtain extensive discussion on the alternative approaches for identifying the nature of local returns to scale at any given input-output bundle on the frontier of the production possibility set. See also Ray (2004; Chapter 3) for an exposition of the alternative approaches and their equivalence.

¹³ Note that the nature of returns to scale of the chosen firms in this fashion will not change if we alter the *orientation* of measurement of efficiency since the production function we consider is globally concave.

4. Description of Variables and Data Used

The present study uses the micro-level data, i.e., the data on a number of variables for different individual industrial units collected by the Central Statistical Organisation (CSO), Government of India through its Annual Survey of Industries (ASI) and made available in soft version. Since these data are not panel data and also quite expensive to procure, we have applied DEA method to obtain individual TE and other related features of production behaviour of the firms for *some* selected years – starting from as early as 1980-81 to 2002-03, but keeping one year gap between two selected adjacent years. In other words we consider 1980-81, 1982-83, 1984-85 and so on till 2002-03 (with the exception for the year 1998-99). We have considered the entire organised leather sector, i.e., the part of the industry for which ASI data are published by CSO on a regular basis.

The *six* variables used in our empirical study are defined below along with their corresponding notations (the notation for the i^{th} firm to be indicated by putting a subscript *i*).

Output (Y): the total ex-factory value¹⁴ of products and byproducts produced by the firm during the year in question.

Intermediate Inputs (*I*): the value of inputs (both indigenous and imported ones, including power, fuels etc.) used by the firm during the year.

Capital (*FA*): the net value of fixed assets¹⁵ of the firm at the beginning of a year.

¹⁴ The definitions of the various concepts like ex-factory value, fixed asset, manday, wages, salaries etc are as used by the CSO. It would have been very useful if we had the panel data over a number of years. However, the lack of sufficient information did not allow one to construct a panel data set from this source.

¹⁵ Ideally one should use Plant and Machinery alone to consider it as an input in a production function. However, we use total value of fixed assets (including value of land and other fixed assets) in view of the observed very high correlation between FA (as we have taken) and value of Plant and Machinery alone. For instance, correlation coefficient between these two variables is as high as 89.85% in the year 2002-03.

Labour (*L*): the total number of mandays worked during the year.

Skill Ratio: the ratio of wages and salaries paid to the managerial and supervisory staff to total wages and salaries paid to all workers and employees.¹⁶

Age: the difference (in years) between the firm's current and initial year of production.

5. Empirical Findings

Using the computer programme DEAP **2.1**, both output-oriented and input-oriented technical efficiency (TE) and scale efficiency (SE) have been calculated, following the DEA methodology, for each individual firm for each sample year relative to the frontier constructed on the basis of the data on all firms in the industry. Subsequently, an (arithmetic) average of these TE and SE values is computed over all firms for each year. These averages are given in the Table 1. The table shows that there is an overall increasing trend in average TE, for both output-oriented and input-oriented measurements. However, there is some downward drift in this trend. From the National Accounts Statistics released by the CSO we also see some temporary downward tendency in the growth rate of GDP of the leather and fur products group started in these years.¹⁷This may be corroborated with the overall performance

We define skill ratio in this manner due to non-availability of data on the suitable variables.

17	For instance	growth r	ate of the	he GDP	of this	group	from	1981-82	through
	$2004\text{-}05 \ \text{are}$	given be	low:						

1981-82	1982-83	1983-84	1984-85	1985-86	1986-87	1987-88	1988-89
22.0	10.8	21.0	16.7	-19.1	-2.5	39.9	3.7
1989-90	1990-91	1991-92	1992-93	1993-94	1994-95	1995-96	1996-97
24.9	17.7	1.2	27.1	50.0	-20.6	-3.7	-10.6
1997-98	1998-99	1999-2000	2000-01	2001-02	2002-03	2003-04	2004-05
34.0	6.9	-8.1	8.3	6.9	-4.2	-4.7	9.0

Source: National Accounts Statistics, CSO, Ministry of Statistics and Programme Implementation, GOI.

of the Indian leather industry for these years. One additional remarkable feature of such drifts is that for the three consecutive sample years since 1990-91 average TE falls gradually, possibly owing to the structural adjustment to the new economic regime. Since we draw separate frontier for each cross-section and, therefore, the benchmark frontier itself changes from one cross-section to another, the improving trend of overall performance seems show only a partial picture and anyone may think that this may be due to the fact that frontier for these three years shift upward due to relatively weaker norms faced by the potential producing units under the new liberalised economic scenario. H/She may also think that this is also true for the two other years namely 1984-85 and 2002-03. But that is clearly not true if we analyse the results shown in Tables 2a and 2b. These tables show that the *gap* between the percentage points showing the firms lies in the IRS segment of the frontier following the input-oriented measurement and at the same time they seem lie in its DRS segment following the outputoriented measurement is wider in most of these five years as compared to the others. But this can happen only when an overwhelming proportion of the firms are much more deviated from the frontier. Thus, the results imply that that is probably the case here for these years and, therefore, their overall efficiency is relatively less. Again, the combined results (of those shown in Table 2a and 2b) following our consideration discussed in Section 3 is shown in Table 2 which fails to show any clear picture about whether the overwhelming majority of the Indian leather firms are operating with too small or too large scale relative to MPSS. They are too small in fifty percent of the sample years and too large in the rest of the cases. Further, overall level of scale efficiency is as high as around 90% for the Indian leather industry as a whole.

Attempts have also been made to find out some explanation for inter-firm differences in such TE's. For instance, some authors (e.g., Aly et al, 1990; Bhandari and Ray, 2007) adopt a two-stage procedure in

this regard and try to find determinants of TE values through regression analyses at the second stage, having estimated these through DEA at the first stage.¹⁸ However, we consider logarithmic value of (output-oriented) TE score (a la Banker et al, 2008) of a firm as the dependent variable while treating firm size, its skill ratio, its age and different group dummies defined below as explanatory variables.¹⁹ The dummy variables are used to examine whether these TE's show any variation across firms in different groups of states, ownership pattern and/or internal organisational structure. We have considered three major leather producing states viz. Tamil Nadu, Uttar Pradesh and West Bengal and incorporating a dummy for each state with firms in all other states being considered as the reference group; a dummy variable for the firms under private ownership with those under public ownership being considered as the reference group and a dummy variable for each of the private limited company (PRLC) and public limited company (PULC) type of internal organisational structure with firms under other types of organisation being considered as the reference group. Further, since skill ratio of a firm, as we have defined, can be considered to be an indicator of the proportion of skilled labourer out of its total labour force, one can expect that it will affect a firm's technical efficiency positively. The estimated results are shown in the Table 3. From the results some clear pictures emerge. First of all, size of a firm affects its TE positively with elasticity of technical efficiency with respect to size (shown in the first row) being statistically significant throughout the years. TE of firms varies significantly with its location and internal

¹⁸ Aly et al (1990) used as determinants of efficiency indices variables like degree of urbanisation, firm's size and product diversity etc. for a sample of 322 independent banks in the USA.

¹⁹ We have also explained the logarithmic value of the input-oriented TE score of the firms using the same set of explanatory variables for the each sample year and the basic results obtained through it (not shown here) remain the same.

organisational structure. To be specific, firms of Tamil Nadu²⁰ are relatively less efficient than the firms located elsewhere and those with two specific organisation types namely private limited company and public limited company are performing worse than their counterpart with other organisational structures. Some other features, which are observed to be relatively less prominent, may be stated as well. The variables like skill ratio and age of firm don't affect its TE significantly in most of the cases. However, whenever their impact becomes significant, its sign is positive. Therefore, there exists some evidence of positive impact of firm's age (during the post-liberalised regime) and a similar impact of skill ratio of a firm on its TE. Similarly, trend regarding the overall effect of ownership pattern of a firm on its TE is insignificant. However, the interesting feature is that whenever it becomes significant it is negative prior to the economic liberalisation and positive in later period which is surely an indicator of the relatively friendly government policy towards the private enterprises after implementation of economic liberalisation on a massive scale in 1991.

An important point about the performance of the firms in West Bengal may be noted. The effect of WB dummy is insignificant in most of the cases and is positive whenever it becomes significant (viz. in 1986-87, 1992-93 and 1999-2000). This clearly shows diametrically opposite picture to the general belief that the overall work culture in West Bengal is far backward than that in many other Indian states. An explanation of their (so-called) better performance can, however, be given in this context. The leather firms in West Bengal are historically clustered mainly in the Beliaghata-Taltala-Entaly-Park Circus-Tangra-Dhapa-Topsia region of Central and East Calcutta, which is located right in the middle of some densely populated areas of the city grown around the tanneries, leather warehouses and workers' slums. There are

²⁰ Tamil Nadu is the largest leather producing state with substantial number of firms located therein for each sample year.

few other fringes as well like Dum Dum, Maniktala, Raja Bazar, New Alipore, Tallygunge, Garia etc. Since tanning, as we have already noted, is the most polluting segment of the production chain of the industry and these firms are not any exception in this regard, in view of their relatively close connection with the residential area, in a verdict the Indian Supreme Court orders them to shift their plants to the newly made Bantala Leather Complex. But most of them did not react accordingly and the political parties, both the alliance in power and the oppositions, are surprisingly silent about them, particularly because they don't want to loose cheap popularity at the cost of the long-run benefit of the mass. The authorised enforcement bodies like the State Government, the Calcutta Municipal Corporation, State Pollution Control Authority, and Department of Police – all are equally reluctant to enforce the Court's order. What happens there? As per the record these firms shut down their operation at the area, but like any other open secret everyone of the surrounding locality knows that these firms continue their operations in full phase while keeping their main gate closed particularly during night. As a result of the entire story these firms are not effectively facing relatively tighter pollution control norms made for the leather industry and, therefore, can be able to incur less (and effectively negligible) pollution abatement cost. So, they become able to use proportionately more resources than an otherwise identical firm located elsewhere in their production activities which may be a factor in showing them to be performing relatively better.

In this regard we can guess why the firms in Tamil Nadu are shown to be relatively less efficient by our analysis despite their operations in the largest leather cluster of the country with the benefit of usual external economies arises from such a clustering. In its Industrial Development Report 2009, the United Nations Industrial Development Organisation (UNIDO) has identified the leather cluster in Chennai, Tamil Nadu among the ten most dynamic industrial location globally for its better environmental practices for those residing in the cluster among some other factors like its competence in international markets, the extent to which it provides better economic and social benefits in terms of employment, skills, wages etc. For their relatively more concern regarding the environmental norms, in particular and labour welfare, in general, (to implement them) these firms have to incur substantial cost due to pollution abatement and more formal practices regarding the labour laws, which may be an important factor for showing them to be less efficient relative to the firms located elsewhere in the country since, even today, almost none of the other clusters in India is doing so well in these dimensions. But the story was not so shining even until few years back²¹ and the Tamil Nadu firms are adjusting themselves relatively faster with the changing overall (sustainable) market conditions than their counterparts elsewhere.

However, as we are unable to construct firm-level panel data from the data set we have and in view of the fact that benchmark frontier, (relative to which individual firms are evaluated), itself changes from one period to another in case of separate cross-sectional analysis for each sample year, our findings of increasing average TE over time may be tentative. However, to overcome this problem, at least to some extent, we construct a representative panel of *average firms*. To be specific, we first arrange the firms in ascending order of their size and club them into ten fractile groups viz. groups containing smallest 10% firms, next 10% firms and so on up to the largest 10% firms and take *only* the average firm (with average value of each variables) from each group. We have done this for each of the sample years. Thus we have a panel of 120 observations with 10 (hypothetical) average firms from each year. Since output, fixed assets and intermediate input variables are measured in

²¹ Interested readers may look at Nihila (1999) to get a detailed picture about how poor the overall working environment was at the Tamil Nadu tanning firms until few years back which is qualitatively no different from that prevails at the leather cluster of Dharavi slum, Mumbai during more recent years (Singh, 2007).

(nominal) value terms, we adjust these figures by the suitable price indices. For instance, output is adjusted by the wholesale price index of manufactured products, fixed asset is adjusted by the wholesale price index of machinery and machine tools group and intermediate input is adjusted by the (simple geometric) mean of the wholesale price indices of non-food primary articles and fuel, power, light and lubricants group. We then estimate a time-varying TE model²² (which is, as we have considered, a panel data stochastic frontier model to estimate a Cobb-Douglas production function with the same three inputs in case of our DEA-based analysis and the same set of explanatory variables (without the dummies) are used to estimate the inefficiency sub-model which we already used in our earlier second-stage regression) with this panel data and obtain individual TE for each hypothetical firm and that for each year. (The detailed results are shown in the Appendix.) The overall level of average TE of these 120 observations is 56% with average level of TE for the individual years 1980-81 through 2002-03 being 46, 49, 53, 52, 59, 55, 56, 62, 59, 56, 57 and 72 percentage points respectively, which clearly shows an increasing trend over time. Therefore, our finding of improving overall performance of the Indian leather manufacturing firms over time is robust in some sense.

6. Concluding Remarks

The leather industry is being considered as one of the most promising industries of India with excellent prospects for growth and export. The government is particularly interested in its promotion in view of its large potential for generating employment and income, with relatively low inputs of capital. A large part of this industry is in the unorganised sector about which very little systemic information is available. To examine some features of the production behavior of this

²² See Cornwell et al (1990), Kumbhakar (1990), Battese and Coelli (1992) for time-varying stochastic frontier models of panel data. An expository discussion of all these models can also be found in Kumbhakar and Lovell (2000; Chapter 3).

industry we have, therefore, no other alternative but to consider the organised part of this industry – the part on which information is available regularly from the Annual Survey of Industries (ASI). In fact, the unitlevel data on industrial firms collected and compiled officially under the ASI are quite rich in coverage and content, but have remained largely unexploited till date.

The industry – at least the part covered by the ASI – grew very fast in the last two decades despite some erratic behaviour of its production in the 1990s. To secure a reasonable position in the export market the industry needs to be efficient in production. The purpose of the present study is to examine the extent of technical efficiency (TE) prevailing among the Indian leather firms as well as behaviour of such efficiency across time and across different groups of firms applying the *data envelopment analysis* (DEA). We have considered firm-level data for *twelve* sample years spanning almost equally around the initiation of economic reforms on a large scale in India in the early 1990s. The main results of our study may be summarised as follows.

Size of a firm affects positively its TE with this effect being statistically significant throughout the years. TE of firms varies significantly with its location and internal organisational structures. To be specific, firms of Tamil Nadu are relatively less efficient than the firms located elsewhere and those with two specific organisation types namely private limited company and public limited company are performing worse than their counterparts with other organisational structures. Some other features, which are observed to be relatively less prominent, may be stated as well. The variables like skill ratio and age of firm don't affect its TE significantly in most of the cases. However, whenever their impact becomes significant, its sign is positive. Therefore, there exists some evidence of positive impact of firm's age (during the post-liberalised regime) and a similar impact of skill ratio of a firm on its TE. Similarly, trend regarding the overall effect of ownership pattern of a firm on its TE is insignificant. However, the interesting feature about it is that whenever it becomes significant it is negative prior to the economic liberalisation and positive in later period which is surely an indicator of the relatively more friendly government policy towards the private enterprises after implementation of economic liberalisation on a massive scale in the early 1990s. Again the effect of WB dummy is insignificant in most of the cases and is positive wherever it becomes significant. Further, on a totality, performance of the Indian leather firms is on an improving trend over time (despite few downswings in it) and overall level of scale efficiency is as high as around 90% for the Indian leather industry as a whole.

Finally, we may propose some policy prescriptions on the basis of our results. As we have already discussed, the policy makers face a dilemma from the demand side regarding whether the Indian leather industry is to be expanded at the face of on-going global economic slowdown. On the other hand, analysing supply side factors we may conclude that there should not be any such hesitation in expanding the industry. If it were such that most of the firms are in the diminishing returns path of the production frontier, then the dilemma definitely would arise from the supply side also. In that case, firms are to contract their scale of operation in order to attain most productive scale size. On the other hand, the observed results clearly show that there exists a significant positive association between firm size and its technical efficiency. As an obvious outcome, the result would be in an ambiguity in taking decision in favour of or against the expansion of the industry. But since no clear picture emerges about the scale of operation of the majority of the firms in different years relative to the respective MPSS, the policy makers' objective should be in favour of expanding the industry, with providing specific emphasis on expanding the size of individual production units, in view of the facts that (a) it is a labour intensive industry with relatively less capital, (which is surely a relatively more scarce resource in India), requirement; (b) India have

huge volume of traditionally skilled cheap labour; (c) unemployment problem in India cannot be ignored for its relatively large share of total labour force and (d) this problem (of unemployment) becomes more severe nowadays as an obvious outcome of the on-going global economic downturn since a massive proportion of her labour force looses their job which comprises both skilled as well as unskilled labour. Of course, all these points are related to the broad issue of problem of unemployment in India. After all, we are observing some sign in the international market of weakening of the ongoing global downturn. According to Mr. Habib Hussain, Chairman, Council for Leather Exports, India, the Indian leather industry expects to end 2009-10 with a 10% decline in export, slightly better relative to the earlier assessment of a 15% decline. He said that there has been a pick up in demand from the markets in the West with factories getting a healthy order inflow from December, 2009 onwards, which will help bring down the decline in growth. The industry is on the path of recovery from the slowdown of 2008 (Business Line, January 31, 2010). In the last quarter (October-December, 2009), the Indian leather industry finally began to see positive growth figures after 12 months of negative performance, Mr. Hussain said (Business Line, February 1, 2010). Government may also encourage the preferred organisational structures for the potential new entrant into the industry. Over and above all these Government should help the existing industrial units turning them into modern-day quality leather producers with advanced technology embodied machinery, better managerial skills, technically trained manpower etc to meet the overall demand of the changing markets.

One obvious limitation of our study is that it assesses possible impact of the ongoing global economic slowdown on the Indian leather industry without analysing data pertaining to the recession years since we have analysed data only upto 2002-03. However, we lay our faith on the long-run historical (supply-side) behaviour of the industry in this regard since we have analysed the relevant data for the last two decades. Again, we use DEA method which is based on the mathematical programming techniques without considering the possible error structures that may affect the analysis. Since each methodology is having its relative advantages as well as disadvantages over any of its possible alternatives, our analysis is not free from its respective limitations.

> Anup Kumar Bhandari is Assistant Professor at the Centre for Development Studies, Thiruvananthapuram. His research interests include Production Economics and related issues like Efficiency and Productivity Analysis; Applied Industrial Economics; Issues related to Indian Banking.

Email: anup@cds.ac.in anupkbhandari@gmail.com

ble 1:	Average	Level of 1	Efficiency	of the Ind	ian Leat	her Firm	S					
	1980-	1982-	1984-	1986-	1988-	1990-	1992-	1994-	-9661	-6661	2000-	2002-
	1981	1983	1985	1987	1989	1991	1993	1995	1997	2000	2001	2003
ο _щ	0.64	0.69	0.42	0.54	0.59	0.52	0.43	0.38	0.53	0.55	0.70	0.64
Qщ	0.65	0.70	0.41	0.54	0.59	0.52	0.45	0.36	0.52	0.56	0.70	0.63
-0 cale	0.96	0.93	0.84	0.82	0.91	0.93	0.92	0.82	0.89	0.93	0.96	0.93
-0 cale	0.93	06.0	0.82	0.77	0.89	0.91	0.83	0.83	0.89	0.90	0.95	0.93
otal rms	362	405	413	468	467	533	567	562	592	321	435	522
	0-0:0	utput-ori	ented; I-C	D: Input-of	"iented;	TE: Tech	inical Ef	ficiency,	Scale:	Scale Eff	ficiency.	

m
ea
Σ
ed
ent
Ē
0
In.
Ith
9
er
nti
Ľ0
еF
th
ng Bu
alo
th
Pa
Ľ
ŝtu
R
int
ere
iff
D
sii
Lie
l St
in.
E
e oi
ag
ent
rce
Pe
a:
le 2
ab.
H

	-	~				<u></u>		
()	2002	200	34	9	60	Scale		
Measure	2000-	2001	41	19	40	eturns to	l points.	
riented	1999-	2000	54	18	28	istant Re	decima	
Dutput-C	-9661	1997	22	7	70	RS: Cor	to TWO	
ontier (C	1994-	1995	22	18	60	le and C	ation up	
ng the Fr	1992-	1993	57	15	29	is to Sca	oproxim	
Path alor	1990-	1991	9	5	89	g Returi	due to al	
[Return]	1988-	1989	40	12	49	minishin	o to 100 .	
n Different	1986-	1987	27	10	63	DRS: Di	not add uj	
ms Lies ir.	1984-	1985	40	6	51	s to Scale	umn may	
age of Fir	1982-	1983	6	5	86	g Return.	in a coli	
: Percent	1980-	1981	49	17	34	ncreasin	Figures	
ľable 2a			IRS	Local CRS	DRS	IRS: I		

eather Firms
nĽ
India
the
of
of Efficiency
Level
Average
÷
þľ

~
Ĺ.
ns
ea
Σ
ed
nt
ie
ō
÷
Idi
E
er
Ţ.
10 <u>1</u>
£
he
a t
on
al
th
Pa
E
tu
Re
nt.
rei
ffe
Dij
[I]
ŝ
Ĕ
S
E
E
$\mathbf{0f}$
ge
ıta
Cen
erc
P
;q
le 2
ldı
Ë

2002-	2003	45	5	50	Scale.	
2000-	2001	50	18	31	turns to	points.
1999-	2000	69	13	18	istant Re	decimal
1996-	1997	42	7	50	RS: Con	to TWO
1994-	1995	59	15	25	le and C	ation up
1992-	1993	77	8	14	is to Sca	pproxim
1990-	1661	19	5	76	g Retury	due to al
1988-	1989	60	6	31	minishin	o to 100
1986-	1987	53	5	43	DRS: Di	not add u
1984-	1985	72	7	21	s to Scale	umn may
1982-	1983	18	4	78	g Return.	in a coli
1980-	1981	99	12	23	ncreasin	Figures
		IRS	Local CRS	DRS	IRS: In	

Table 2: Percentage of Firms Lies in Different Return Path along the Frontier (Combined Results)

2002-	2003	34	50	Scale.
2000-	2001	41	31	turns to
1999-	2000	54	18	istant Re
1996-	1997	22	50	RS: Con
1994-	1995	22	25	le and C
1992-	1993	57	14	ts to Sca
1990-	1661	9	76	g Return
1988-	1989	40	31	minishin
1986-	1987	27	43	DRS: Di
1984-	1985	40	21	s to Scale,
1982-	1983	6	78	g Return
1980-	1981	49	23	ncreasin
		IRS	DRS	IRS: In

Table 3: Estimated Coefficients of the Explanatory Variables to Explain Logarithmic Value of (Output-Oriented) **Technical Efficiency**

(38) 1983 1985 1987 1983 1985 1987 1983 1985 1987 1983 1985 1987 1983 1985 1987 1981 1985 1981 1985 1981 1985 1981 1985 1981 1983 1983 1983 1983 1031 0.13 0.13 0.25 0.02 0.03 <t< th=""><th>atory</th><th>1980-</th><th>1982-</th><th>1984-</th><th>1986-</th><th>1988-</th><th>1990-</th><th>1992-</th><th>1994-</th><th>1996-</th><th>1999-</th><th>2000-</th><th>2002-</th></t<>	atory	1980-	1982-	1984-	1986-	1988-	1990-	1992-	1994-	1996-	1999-	2000-	2002-
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1	981	1983	1985	1987	1989	1661	1993	1995	1997	2000	2001	2003
(2.4) (4.3) (7.7) (9.4) (7.6) (6.1) (5.7) (4.7) (8.8) (4.8) (4.8) (6.8) 0.27 -0.04 -0.71 0.26 -0.08 -0.02 -0.44 0.11 0.30 0.05 </td <td></td> <td>0.06</td> <td>0.09</td> <td>0.23</td> <td>0.30</td> <td>0.18</td> <td>0.13</td> <td>0.25</td> <td>0.20</td> <td>0.25</td> <td>0.10</td> <td>0.11</td> <td>0.13</td>		0.06	0.09	0.23	0.30	0.18	0.13	0.25	0.20	0.25	0.10	0.11	0.13
0.27 -0.04 0.01 0.26 -0.08 -0.02 -0.44 0.11 0.30 0.05 0.05 0.07 (0.9) (-0.2) (2.7) (1.3) (-0.3) (-0.1) (-1.8) (0.7) (1.6) (0.3) (0.5) (1.0) (0.7) (1.3) (-1.4) (1.0) 0.00 0.00 0.00 (0.3) (2.4) (1.0) (0.7) (1.3) (-1.4) (1.0) 0.03 (0.3) (-2.0) (-2.0) (-2.0) (1.0) (0.7) (1.3) (-1.4) (1.0) (0.8) (-3.5) (-4.3) (-3.7) (-1.3) (-2.1) (-1.3) (-2.1) (-1.0) (-1.0) (-1.0) (-1.0) (-1.0) (-1.0) (-1.0) (-1.0) (-1.0) (-0.1) (-1.0) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9) (-0.9		(2.4)	(4.3)	(7.7)	(9.4)	(2.6)	(6.1)	(5.7)	(4.7)	(8.8)	(4.8)	(4.8)	(6.8)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		0.27	-0.04	-0.04	0.71	0.26	-0.08	-0.02	-0.44	0.11	0.30	0.05	0.07
0.00 0.00 </td <td></td> <td>(0.9)</td> <td>(-0.2)</td> <td>(-0.2)</td> <td>(2.7)</td> <td>(1.3)</td> <td>(-0.3)</td> <td>(-0.1)</td> <td>(-1.8)</td> <td>(0.7)</td> <td>(1.6)</td> <td>(0.3)</td> <td>(0.5)</td>		(0.9)	(-0.2)	(-0.2)	(2.7)	(1.3)	(-0.3)	(-0.1)	(-1.8)	(0.7)	(1.6)	(0.3)	(0.5)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		0.00	0.00	00.00	-0.00	00.00	0.00	-0.00	-0.00	0.00	0.00	0.00	0.00
-0.42 -0.42 -0.42 -0.42 -0.56 -0.33 -0.34 -0.37 -0.50 -0.34 -0.11 -0.24 -0.15 (-3.5) (-4.4) (-3.5) (-4.3) (-3.5) (-4.3) (-5.2) (-4.2) (-1.3) (-2.0) -0.14 -0.15 (-1.5) (-1.7) (-1.3) (0.6) (-1.0) (-1.0) (-0.6) (0.3) (-0.9) -0.19 -0.24 0.03 0.03 (-0.3) (-0.2) (-0.9) (-0.9) -1.10 (-1.5) (-1.7) (-1.1) (-1.1) (-1.1) (-1.0) (-0.9) (-0.9) -0.11 (-2.3) (0.3) (-0.3) (-0.2) (-1.1) (-1.2) (-1.2) (-1.2) (-1.2) (-1.2) (-0.93) (-0.91) (-0.92) (-0.16) (-1.2) (-1.2) (-1.2) (-1.2) (-1.2) (-1.9) (-1.9) (-1.9) (-1.9) (-1.9) <td< td=""><td></td><td>(1.0)</td><td>(0.7)</td><td>(1.3)</td><td>(-1.4)</td><td>(1.0)</td><td>(0.8)</td><td>(-0.5)</td><td>(-0.2)</td><td>(2.0)</td><td>(0.9)</td><td>(0.8)</td><td>(2.4)</td></td<>		(1.0)	(0.7)	(1.3)	(-1.4)	(1.0)	(0.8)	(-0.5)	(-0.2)	(2.0)	(0.9)	(0.8)	(2.4)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		-0.4	-0.42	-0.42	-0.56	-0.33	-0.34	-0.37	-0.50	-0.34	-0.11	-0.24	-0.15
-0.04 -0.15 -0.17 -0.15 0.05 -0.10 -0.11 -0.17 -0.06 0.01 -0.06 -0.19 -0.24 (-1.5) (-1.7) (-1.3) (0.6) (-1.0) (-1.6) (-1.0) (-0.6) (0.3) (-0.9) -0.19 -0.24 0.03 0.03 0.03 (-0.3) (-1.2) (-1.1) (+1.2) (-1.1) (+1.2) (-0.1) (-0.6) (0.3) (-0.9) (-1.1) (-1.6) (0.3) (0.3) (0.3) (0.3) (-0.3) (-1.1) (+1.2) (-1.1) (+1.2) (-1.1) (+1.2) (-1.1) (+1.2) (0.1) (11.1) (0.71) (0.22) -0.16 -0.22 -0.23 -0.24 -0.12 -0.23 -0.23 -0.20 (1.1) (0.71) (0.22) (-1.8) (-0.6) (-2.3) (-1.8) (-2.0) (-1.3) (-1.3) (-1.3) (-1.3) (-1.3) (-1.3) (-1.3) (-1.3) (-1.3) <th< td=""><td></td><td>(-3.5)</td><td>(-4.4)</td><td>(-3.5)</td><td>(-4.6)</td><td>(-3.5)</td><td>(-4.3)</td><td>(-3.7)</td><td>(-5.2)</td><td>(-4.2)</td><td>(-1.3)</td><td>(-2.6)</td><td>(-2.0)</td></th<>		(-3.5)	(-4.4)	(-3.5)	(-4.6)	(-3.5)	(-4.3)	(-3.7)	(-5.2)	(-4.2)	(-1.3)	(-2.6)	(-2.0)
		-0.04	-0.15	-0.17	-0.15	0.05	-0.10	-0.11	-0.17	-0.07	-0.06	0.01	-0.04
-0.19 -0.24 0.03 0.33 -0.03 0.34 0.16 -0.16 0.34 0.01 0.03 (-1.1) (-1.6) (0.3) (2.3) (0.3) (-0.3) (2.3) (0.1) (1.1) (4.2) (0.1) (1.1) 0.34 0.13 0.02 -0.26 -0.11 -0.22 -0.22 -0.14 0.64 -0.22 0.18 (1.1) (0.7) (0.2) (-1.7) (-0.9) (-2.8) (-1.7) (-0.9) (-2.8) (-1.7) (-0.9) (-2.9) (-1.3) (1.8) 0.05 -0.24 -0.16 -0.23 -0.21 (-1.3) (-1.3) (-1.3) 0.01 (0.2) (-1.18) (-0.6) (-3.8) (-4.8) (-2.6) (-1.8) (-2.0) 0.02 -0.16 (0.44) (-1.4) (-0.8) (-1.2) (-1.9) (-2.9) (-1.9) (-2.0) (-1.9) (-2.0) (-1.9) (-2.0) (-1.9) (-2.9) (-1.9) <t< td=""><td></td><td>(-0.4)</td><td>(-1.5)</td><td>(-1.7)</td><td>(-1.3)</td><td>(0.6)</td><td>(-1.0)</td><td>(-1.0)</td><td>(-1.6)</td><td>(0.1-)</td><td>(9.0-)</td><td>(0.3)</td><td>(6.0-)</td></t<>		(-0.4)	(-1.5)	(-1.7)	(-1.3)	(0.6)	(-1.0)	(-1.0)	(-1.6)	(0.1-)	(9.0-)	(0.3)	(6.0-)
		-0.19	-0.24	0.03	0.30	0.03	-0.03	0.34	0.16	-0.16	0.34	0.01	0.05
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(-1.1)	(-1.6)	(0.3)	(2.3)	(0.3)	(-0.3)	(2.4)	(1.2)	(-1.1)	(4.2)	(0.1)	(1.1)
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		0.34	0.13	0.02	-0.26	-0.11	-0.22	-0.22	-0.69	-0.14	0.64	-0.22	0.18
-0.05 -0.20 -1.0 -0.24 -0.16 -0.05 -0.34 -0.45 -0.21 -0.29 -0.20 -0.09 (-0.4) (-1.4) (-0.8) (-2.0) (-1.8) (-0.6) (-3.8) (-4.8) (-2.6) (-3.8) (-1.8) (-2.0) (-0.3) -0.16 0.05 -0.24 -0.16 0.047 -0.37 (-2.0) (-1.9) (-2.0) (-1.9) (-2.0) (-1.9) (-2.0) (-1.9) (-2.1) (-1.2) (-1.9) (-2.1) (-1.2) (-1.9) (-2.1) (-1.2) (-1.9) (-2.0) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-1.9) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) (-2.1) $(-2.1$		(1.1)	(0.7)	(0.2)	(-1.7)	(6.0-)	(-2.8)	(-0.8)	(-1.7)	(-0.4)	(2.4)	(-1.3)	(1.8)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		-0.05	-0.20	-1.0	-0.24	-0.16	-0.05	-0.34	-0.45	-0.21	-0.29	-0.20	-0.09
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(-0.4)	(-1.4)	(-0.8)	(-2.0)	(-1.8)	(9.0-)	(-3.8)	(-4.8)	(-2.6)	(-3.8)	(-1.8)	(-2.0)
(0.2) (-1.0) (0.4) (-2.7) (-1.5) (-1.2) (-3.8) (-3.3) (-3.6) (-2.3) (-1.3) (-1.9) -1.83 -1.74 -4.47 -4.96 -3.46 -2.60 -4.63 -3.52 -4.83 -3.12 -2.08 -2.87 (-4.3) (-5.6) (-9.7) (-10.3) (-8.5) (-7.7) (-5.7) (-3.8) (-7.2) (-5.0) (-8.8) (-4.3) (-5.6) (-9.7) (-10.3) (-8.5) (-7.7) (-5.7) (-3.8) (-7.2) (-5.0) (-8.8) (-4.3) (-5.6) (-9.7) (-10.3) (-8.5) (-7.7) (-3.8) (-7.2) (-5.0) (-8.8) 3.1 4.2 14.6 22.0 14.3 8.2 17.3 19.4 23.8 10.4 8.2 21.5 562 405 457 533 567 562 592 321 435 522 Figure in parenthesis is corresponding (heteroscedasticity-adjusted,		0.03	-0.16	0.05	-0.36	-0.24	-0.16	-0.47	-0.37	-0.34	-0.32	-0.08	-0.22
-1.83 -1.74 -4.47 -4.96 -3.46 -2.60 -4.63 -3.52 -4.83 -3.12 -2.08 -2.87 (-4.3) (-5.6) (-9.7) (-10.3) (-8.5) (-7.7) (-3.8) (-8.6) (-7.2) (-5.0) (-8.8) 3.1 4.2 14.6 22.0 14.3 8.2 17.3 19.4 23.8 10.4 8.2 21.5 362 405 413 468 467 533 567 562 592 321 435 522 Figure in parenthesis is corresponding (heteroscedasticity-adjusted, a la White) t-ratio. snot contaminating our estimation. snot contaminating our estimation. as 0.0 and 0.00 are very small and are approximated up to one and two decimal points respectively. and to one and two decimal points respectively.		(0.2)	(-1.0)	(0.4)	(-2.7)	(-1.5)	(-1.2)	(-3.8)	(-3.3)	(-3.6)	(-2.3)	(-1.3)	(6.1-)
(-4.3) (-5.6) (-9.7) (-10.3) (-8.5) (-7.7) (-3.8) (-8.6) (-7.2) (-5.0) (-8.8) 3.1 4.2 14.6 22.0 14.3 8.2 17.3 19.4 23.8 10.4 8.2 21.5 362 405 413 468 467 533 567 562 592 321 435 522 Figure in parenthesis is corresponding (heteroscedasticity-adjusted, a la White) t-ratio. and 0.00 are very stand and are annototimeted with the other contaminating our estimation.		-1.83	-1.74	-4.47	-4.96	-3.46	-2.60	-4.63	-3.52	-4.83	-3.12	-2.08	-2.87
3.1 4.2 14.6 22.0 14.3 8.2 17.3 19.4 23.8 10.4 8.2 21.5 362 405 413 468 467 533 567 562 592 321 435 522 Figure in parenthesis is corresponding (heteroscedasticity-adjusted, a la White) t-ratio.w as 0.0 and 0.00 are very small and are approximated up to one and two decimal points respectively.		(-4.3)	(-5.6)	(7.6-)	(-10.3)	(-8.5)	(-7.7)	(-5.7)	(-3.8)	(-8.6)	(-7.2)	(-5.0)	(-8.8)
362 405 413 468 467 533 567 562 592 321 435 522 Figure in parenthesis is corresponding (heteroscedasticity-adjusted, a la White) t-ratio. idual standard errors are not too large multicollinearity, if at all, is not contaminating our estimation. sepactively.		3.1	4.2	14.6	22.0	14.3	8.2	17.3	19.4	23.8	10.4	8.2	21.5
Figure in parenthesis is corresponding (heteroscedasticity-adjusted, a la White) t-ratio. idual standard errors are not too large multicollinearity, if at all, is not contaminating our estimation. wn as 0.0 and 0.00 are very small and are approximated up to one and two decimal points respectively.		362	405	413	468	467	533	567	562	592	321	435	522
idual standard errors are not too large multicollinearity, if at all, is not contaminating our estimation. wn as 0.0 and 0.00 are very small and are approximated up to one and two decimal points respectively.		Figure	e in parei	athesis is	correspon	ding (he	terosced	asticity-aq	ljusted, a l	a White) t	-ratio.		
wn as 0.0 and 0.00 are verv small and are approximated in to one and two decimal noints respectively.		vidual st	andard e	rrors are	not too la	"ge multi	icollinea	rity, if at a	ill, is not c	ontamina	ting our e	estimatio	'n.
	- 67	0 so um	0 and 0 1	10 are ver	v small an	id are an	nroxima	ted un to c	ne and tw	o decimal	noints r	esnective	l_{V}

Appendix

Table A.1: Estimated Cobb-Douglas Stochastic Frontier Production
Function with Hypothetical (Average) Panel Data

Variable	Parameter	Estimated Value of the Parameter*		
Constant	β_0	2.63 (1.93)		
in I	β_1	0.67 (9.40)		
ln FA	β_2	0.25 (8.43)		
ln L	β_3	- 0.03 (- 0.58)		
Constant	δ_0	3.97 (2.96)		
ln I	δ_1	- 0.20 (- 2.87)		
Age	δ_2	- 0.01 (- 2.07)		
Skill Ratio δ_3		0.17 (0.42)		
$\sigma^2 = \sigma_i$	$\left(\frac{2}{4}+\sigma_v^2\right)$	0.02 (6.99)		
$\gamma \left(=\sigma_u^2/\sigma^2\right)$		0.89 (7.73)		
Log-Likelih	ood Value	62.00		
No. Of	Firms	10		
No. of Tin	ne Period	12		
Total No. of	Observation	120		
Overall Me	an TE (%)	56.24		
* Figu	re in the parent	hesis is corresponding t-ratio.		

Table A.2: Individual TE Score of the Hypothetical (Average) Firms in Different Years

Year	2002	-03	0.33	0.52	0.57	0.65	0.70	0.76	0.81	0.93	0.92	0.98	0.72
	2000-	01	0.27	0.36	0.42	0.43	0.50	0.58	0.67	0.73	0.76	0.96	0.57
	1999-	2000	0.31	0.36	0.47	0.51	0.53	0.44	0.59	0.66	0.82	0.90	0.56
	1996-	76	0.34	0.51	0.46	0.41	0.58	0.60	0.77	0.68	0.79	0.79	0.59
	1994	-95	0.46	0.45	0.47	0.41	0.54	0.61	0.74	0.73	0.85	0.92	0.62
	1992	-93	0.29	0.35	0.36	0.43	0.51	0.56	0.73	0.68	0.75	06.0	0.56
	1990-	16	0.23	0.32	0.37	0.44	0.50	0.61	0.65	0.70	0.73	0.95	0.55
	1988-	89	0.33	0.41	0.39	0.47	0.52	0.62	0.62	0.74	0.86	0.96	0.59
	1986-	87	0.22	0.31	0.38	0.34	0.47	0.51	0.64	0.69	0.69	0.93	0.52
	1984-	85	0.25	0.40	0.40	0.35	0.59	0.50	0.53	0.66	0.67	0.94	0.53
	1982-	83	0.29	0.33	0.35	0.38	0.34	0.52	0.55	0.60	0.62	0.90	0.49
	1980-	81	0.27	0.32	0.32	0.38	0.45	0.39	0.52	0.55	0.57	0.82	0.46
Fractile	Group			2	3	4	5	9	7	8	6	10	Average

35

References

- Aly, H.Y., R. Grabowski, C. Pasurka, and N. Rangan (1990), "Technical, Scale, and Allocative Efficiencies in U. S. Banking: An Empirical Investigation" *Review of Economics and Statistics* 72 (2): 211-218.
- Banerjee, N., and M. Nihila (1999), "Business Organization in the Leather Industry of Calcutta and Chennai", In A.K. Bagchi (Ed) *Economy* and Organization: Indian Institutions under the Neoliberal Regime, Sage Publications: New Delhi.
- Banker, R.D. (1984), "Estimating the Most Productive Scale Size Using Data Envelopment Analysis", *European Journal of Operational Research* 17 (1): 35-44
- Banker, R.D., A. Charnes, and W.W. Cooper (1984), "Some Models for Estimating Technical and Scale Inefficiencies in Data Envelopment Analysis", *Management Science* 30 (9): 1078-1092.
- Banker, R.D., and R. Natarajan (2008), "Evaluating Contextual Variables Affecting Productivity Using Data Envelopment Analysis", *Operations Research* 56 (1): 48-58.
- Battese, G.E., and T.J. Coelli (1992), "Frontier Production Functions, Technical Efficiency and Panel Data: With Application to Paddy Farmers in India", *Journal of Productivity Analysis* 3 (1/2): 153-169.
- Bhandari, A.K. (2009), "Selected Aspects of Performance of Indian Industries: An Empirical Investigation," Unpublished Ph.D. Thesis (Chapter 6).
- Bhandari, A.K., and P. Maiti (2007), "Efficiency of Indian Manufacturing Firms: Textile Industry as a Case Study," *International Journal* of Business and Economics 6(1):71-88.
- Bhandari, A.K., and S.C. Ray (2007), "Technical Efficiency in the Indian Textiles Industry: A Nonparametric Analysis of Firm-Level Data", *Department of Economics Working Paper Series*: 49, University of Connecticut.

- Bhavani, T.A. (1991), "Technical Efficiency in Indian Modern Small Scale Sector: An Application of Frontier Production Function", *Indian Economic Review* 26 (2): 149-166.
- Charnes, A., W.W. Cooper, and E. Rhodes (1978), "Measuring the Efficiency of Decision Making Units," *European Journal of Operational Research* 2 (6): 429-444.
- Cornwell, C., P. Schmidt, and R.C. Sickles (1990), "Production Frontiers with Cross-Sectional and Time-Series Variation in Efficiency Levels", *Journal of Econometrics* 46 (1/2): 185-200.
- Färe, R., S. Grosskopf, and C.A.K. Lovell (1985), *The Measurement of Efficiency of Production*, Kluwer-Nijhoff: Boston.
- Goldar, B. (1985), "Unit Size and Economic Efficiency in Small ScaleWashing Soap Industry in India", *Artha Vijnana* 27 (1): 21-40.
- Goldar, B., V.S. Renganathan, and R. Banga (2004), "Ownership and Efficiency in Engineering Firms: 1990-91 to 1999-2000", *Economic and Political Weekly* 39 (5): 441-447.
- Government of India (2004-05), Economic Survey, Ministry of Finance.
- Kumbhakar, S.C. (1990), "Production Frontiers, Panel Data, and Time-Varying Technical Inefficiency", *Journal of Econometrics* 46 (1/2): 201-212.
- Kumbhakar, S.C., and C.A.K. Lovell (2000), *Stochastic Frontier Analysis*, Cambridge University Press: Cambridge.
- Lall, S.V., and G.C. Rodrigo (2001), "Perspectives on the Sources of Heterogeneity in Indian Industry", World Development 29 (12): 2127-2143.
- Little, I.M.D., D. Mazumdar, and J.M. Page Jr. (1987), Small Manufacturing Enterprises: A Comparative Analysis of India and Other Economies, Oxford University Press: Oxford.

- Marshall, A. (1920), *Principles of Economics* (8th Edition), MacMillan: London.
- Mohapatra, K.M., and K. Srivastava (2002), "Leather Goods Industry in Kanpur: Casteism, Religion and Business Interlinkings", *Economic and Political Weekly* 37 (39): 4029-4035.
- Mukherjee, K., and S.C. Ray (2004), "Technical Efficiency and its Dynamics in Indian Manufacturing: An Inter-State Analysis", *Department of Economics Working Paper Series*: 18, University of Connecticut.
- Neogi, C., and B. Ghosh (1994), "Intertemporal Efficiency Variations in Indian Manufacturing Industries", *Journal of Productivity Analysis* 5 (3): 301-324.
- Nihila, M. (1999), "Marginalisation of Women Workers: Leather Tanning Industry in Tamil Nadu", *Economic and Political Weekly*, 34 (16/ 17), WS21-WS27.
- Nikaido, Y. (2004), "Technical Efficiency of Small-Scale Industry: Application of Stochastic Production Frontier Model", *Economic* and Political Weekly 39 (6): 592-597.
- Page, Jr. J.M. (1984), "Firm Size and Technical Efficiency: Application of Production Frontiers to Indian Survey Data", *Journal of Development Economics* 16 (1/2): 129-152.
- Penrose, E.T. (1959), *The Theory of the Growth of the Firm*, Basil Blackwell: Oxford.
- Ramaswamy, V.K. (1994), "Technical Efficiency in Modern Small-Scale Firms in Indian Industry: Applications of Stochastic Production Frontier", *Journal of Quantitative Economics* 10 (2): 309-324.
- Ray, S.C. (2004), *Data Envelopment Analysis: Theory and Techniques* for Economics and Operations Research, Cambridge University Press: Cambridge.

- Singh, A. (2007), "Strategies for Enhancing the Competitiveness of Leather Industry in India", Paper presented at the Conference on Global Competition & Competitiveness of Indian Corporates, Indian Institute of Management, Kozhikode, May 18-20.
- Stinchcombe, A.L. (1965), "Social Structure and Organizations", In J.G. March (Ed), Handbook of Organizations, Rand McNally: Chicago.

PUBLICATIONS

For information on all publications, please visit the CDS Website: www.cds.edu. The Working Paper Series was initiated in 1971. Working Papers from 279 can be downloaded from the site.

The Working Papers published after April 2007 are listed below:

- W.P. 425 P L. BEENA, HRUSHIKESH MALLICK, Exchange Rate and Export Behaviour of Indian Textiles & Clothing Sector: An Enquiry for Major Destination Countries. March 2010.
- W.P. 424 K. C. ZACHARIAH, S. IRUDAYA RAJAN, Migration Monitoring Study, 2008 Emigration and Remittances in the Context of Surge in Oil Prices. March 2010.
- W.P. 423 VIJAYAMOHANAN PILLAI N Loss of Load Probability of a Power System: Kerala. February 2010.
- W.P. 422 JAYASEKHAR S, C. NALIN KUMAR, Compliance, Competitiveness and Market Access: A Study on Indian Seafood Industry. February 2010.
- W.P. 421 S. IRUDAYA RAJAN, V.J. VARGHESE, M.S. JAYAKUMAR Overseas Recruitment in India: Structures, Practices and Remedies. December 2009.
- W.P. 420 V.J. VARGHESE, Land, Labour and Migrations: Understanding Kerala's Economic Modernity December 2009
- W.P. 419 R.MOHAN, D. SHYJAN Tax Devolution and Grant Distribution to States in India Analysis and Roadmap for Alternatives, December 2009.
- W.P. 418 WILLIAM JOE & U.S. MISHRA Household Out-of-Pocket Healthcare Expenditure in India Levels, Patterns and Policy Concerns, October 2009.
- W.P. 417 NEETHI P Globalisation Lived Locally: New Forms of Control, Conflict and Response Among Labour in Kerala, Examined Through a Labour Geography Lens. October 2009.

- W.P. 416 SUNIL MAN High Skilled Migration from India, An Analysis of its Economic Implications. September 2009.
- W.P. 415 SUNIL MANI Has India Become more Innovative Since 1991? Analysis of the Evidence and Some Disquieting Features, September 2009.
- W.P. 414 WILLIAM JOE, PRIYAJIT SAMAIYAR, U.S. MISHRA Migration and Urban Poverty in India Some Preliminary Observations, September 2009.
- W.P. 413 K.N.NAIR, T.P. SREEDHARAN, M. ANOOPKUMAR, A Study of National Rural Employment Guarantee Programme in Three Grama Panchayats of Kasaragod District, August 2009
- W.P. 412 B.S. SURAN, D. NARAYANA, The Deluge of Debt: Understanding the Financial Needs of Poor Households. July 2009
- W.P. 411 K. NAVANEETHAM, M. KABIR, C.S. KRISHNAKUMAR

Morbidity Patterns in Kerala: Levels and Determinants. April 2009.

- W.P. 410 ARINDAM BANERJEE, Peasant Classes, Farm Incomes and Rural Indebtedness: An Analysis of Household Production Data from two States. March 2009.
- W.P. 409 SUNIL MANI, The Growth of Knowledge-intensive Entrepreneurship in India, 1991-2007 Analysis of its Evidence and the Facilitating Factors. February, 2009
- W.P. 408 M. S. HARILAL, Home to Market: Responses, Resurgence and Transformation of Ayurveda from 1830s to 1920. November 2008
- W.P. 407 HRUSHIKESH MALLICK, Do Remittances Impact the Economy ? Some Empirical Evidences from a Developing Economy. October 2008.
- W.P. 406 K.C.ZACHARIAH, S.IRUDAYA RAJAN, Costs of Basic Services in Kerala, 2007, Education, Health, Childbirth and Finance (Loans) September 2008.
- W.P. 405 SUNIL MANI Financing of industrial innovations in India How effective are tax incentives for R&D? August 2008.

- W.P. 404 VINOJ ABRAHAM Employment Growth in Rural India: Distress Driven? August 2008.
- W.P. 403 HRUSHIKESH MALLICK, Government Spending, Trade Openness and Economic Growth in India: A Time Series Analysis. July 2008.
- W.P. 402 K. PUSHPANGADAN, G. MURUGAN, Dynamics of Rural Water Supply in Coastal Kerala: A Sustainable Development View, June 2008
- W.P. 401 K. K. SUBRAHMANIAN, SYAM PRASAD, Rising Inequality With High Growth Isn't this Trend Worrisome? Analysis of Kerala Experience, June 2008
- W.P. 400 T.R. DILIP, Role of Private Hospitals in Kerala: An Exploration, June 2008
- W.P. 399 V. DHANYA, Liberalisation of Tropical Commodity Market and Adding-up Problem: A Bound Test Approach, March 2008
- W.P. 398 P. MOHANAN PILLAI, N. SHANTA, ICT and Employment Promotion Among Poor Women: How can we Make it Happen? Some Reflections on Kerala's Experience. February 2008.
- W.P. 397 K.N.NAIR, VINEETHA MENON, Distress Debt and Suicides among Agrarian Households: Findings from three Village Studies in Kerala. December 2007
- W.P. 396 K.N.NAIR, C.P. VINOD, VINEETHA MENON, Agrarian Distress and Livelihood Strategies: A Study in Pulpalli Panchayat, Wayanad District, Kerala December 2007
- W.P. 395 K.C. ZACHARIAH, S.IRUDAYA RAJAN, Migration, Remittances And Employment Short-term Trends and Longterm Implications. December 2007
- W.P. 394 K.N.NAIR, ANTONYTO PAUL, VINEETHA MENON, Livelihood Risks and Coping strategies: A Case Study in the Agrarian Village of Cherumad, Kerala. November 2007
- W.P. 393 S. IRUDAYA RAJAN, U.S.MISHRA, Managing Migration in the Philippines: Lessons for India. November 2007.
- W.P. 392 K.N. NAIR, R. RAMAKUMAR Agrarian Distress and Rural Livelihoods, a Study in Upputhara Panchayat Idukki District, Kerala. November 2007.

- W.P. 391 PULAPRE BALAKRISHNAN, Visible hand: Public policy and economic growth in the Nehru era. November 2007.
- W.P. 390 SUNIL MANI, The Growth Performance of India's Telecommunications Services Industry, 1991-2006 Can it Lead to the Emergence of a Domestic Manufacturing Hub? September 2007.
- W.P. 389 K. J. JOSEPH, VINOJ ABRAHAM, Information Technology and Productivity: Evidence from India's Manufacturing Sector. September 2007.
- W.P. 388 HRUSHIKESH MALLICK, Does Energy Consumption Fuel Economic Growth In India? September 2007.
- W.P. 387 D. SHYJAN, Public Investment and Agricultural Productivity: A State-wise Analysis of Foodgrains in India. July 2007.
- W.P. 386 J. DEVIKA, 'A People United in Development': Developmentalism in Modern Malayalee Identity. June 2007.
- W.P. 385 M. PARAMESWARAN, International Trade, R&D Spillovers and Productivity: Evidence from Indian Manufacturing Industry. June 2007.
- W.P. 384 K. C. ZACHARIAH, S. IRUDAYA RAJAN Economic and Social Dynamics of Migration in Kerala, 1999-2004 Analysis of Panel Data. May 2007.
- W.P. 383 SAIKAT SINHA ROY Demand and Supply Factors in the Determination or India's Disaggregated Manufactured Exports : A Simultaneous Error-Correction Approach. May 2007.